

Highly Active Ice-Nucleating Particles at the Summer North Pole

Downloaded from: https://research.chalmers.se, 2022-07-02 09:44 UTC

Citation for the original published paper (version of record):

Porter, G., Adams, M., Brooks, I. et al (2022). Highly Active Ice-Nucleating Particles at the Summer North Pole. Journal of Geophysical Research: Atmospheres, 127(6). http://dx.doi.org/10.1029/2021JD036059

N.B. When citing this work, cite the original published paper.

research.chalmers.se offers the possibility of retrieving research publications produced at Chalmers University of Technology. It covers all kind of research output: articles, dissertations, conference papers, reports etc. since 2004. research.chalmers.se is administrated and maintained by Chalmers Library



JGR Atmospheres

RESEARCH ARTICLE

10.1029/2021JD036059

Key Points:

- The concentration of ice-nucleating particles at the North Pole in summer 2018 was amongst the highest anywhere in the world
- Biological ice-nucleating particles were derived from the Russian seas and perhaps associated with winddriven sea spray
- The concentration of ice-nucleating particles at the surface was often different to that higher in the boundary layer where clouds form

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

B. J. Murray, b.j.murray@leeds.ac.uk

Citation:

Porter, G. C. E., Adams, M. P., Brooks, I. M., Ickes, L., Karlsson, L., Leck, C., et al. (2022). Highly active ice-nucleating particles at the summer North Pole. *Journal of Geophysical Research: Atmospheres, 127*, e2021JD036059. https://doi.org/10.1029/2021JD036059

Received 19 OCT 2021 Accepted 18 FEB 2022

Author Contributions:

Conceptualization: Grace C. E. Porter, Ian M. Brooks, Caroline Leck, Matthew E. Salter, Julia Schmale, Paul Zieger, Benjamin J. Murray Data curation: Grace C. E. Porter, Matthew E. Salter, Paul Zieger, Benjamin J. Murray Formal analysis: Grace C. E. Porter, Ian M. Brooks, Luisa Ickes, Linn Karlsson, Caroline Leck, Julia Schmale, Karolina Siegel, Sebastien N. F. Sikora, Jutta

Vüllers, Heini Wernli, Julika Zinke,

Benjamin J. Murray

© 2022. The Authors. This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Highly Active Ice-Nucleating Particles at the Summer North Pole

Grace C. E. Porter^{1,2}, Michael P. Adams¹, Ian M. Brooks¹, Luisa Ickes³, Linn Karlsson^{4,5}, Caroline Leck^{4,6}, Matthew E. Salter^{4,5}, Julia Schmale⁷, Karolina Siegel^{4,5,6}, Sebastien N. F. Sikora¹, Mark D. Tarn^{1,2}, Jutta Vüllers^{1,8}, Heini Wernli⁹, Paul Zieger^{4,5}, Julika Zinke^{4,5}, and Benjamin J. Murray¹

¹School of Earth and Environment, University of Leeds, Leeds, UK, ²School of Physics and Astronomy, University of Leeds, Leeds, UK, ³Department of Space, Earth and Environment, Chalmers University, Gothenburg, Sweden, ⁴Bolin Centre for Climate Research, Stockholm University, Stockholm, Sweden, ⁵Department of Environmental Science, Stockholm University, Stockholm, Sweden, ⁶Department of Meteorology, Stockholm University, Stockholm, Sweden, ⁷School of Architecture, Civil and Environmental Engineering, École Polytechnique Fédérale de Lausanne, Lausanne, Switzerland, ⁸Now at Institute of Meteorology and Climate Research, Karlsruhe Institute of Technology, Karlsruhe, Germany, ⁹Institute for Atmospheric and Climate Science, ETH Zürich, Zürich, Switzerland

Abstract The amount of ice versus supercooled water in clouds is important for their radiative properties and role in climate feedbacks. Hence, knowledge of the concentration of ice-nucleating particles (INPs) is needed. Generally, the concentrations of INPs are found to be very low in remote marine locations allowing cloud water to persist in a supercooled state. We had expected the concentrations of INPs at the North Pole to be very low given the distance from open ocean and terrestrial sources coupled with effective wet scavenging processes. Here we show that during summer 2018 (August and September) high concentrations of biological INPs (active at $>-20^{\circ}$ C) were sporadically present at the North Pole. In fact, INP concentrations were sometimes as high as those recorded at mid-latitude locations strongly impacted by highly active biological INPs, in strong contrast to the Southern Ocean. Furthermore, using a balloon borne sampler we demonstrated that INP concentrations were often different at the surface versus higher in the boundary layer where clouds form. Back trajectory analysis suggests strong sources of INPs near the Russian coast, possibly associated with wind-driven sea spray production, whereas the pack ice, open leads, and the marginal ice zone were not sources of highly active INPs. These findings suggest that primary ice production, and therefore Arctic climate, is sensitive to transport from locations such as the Russian coast that are already experiencing marked climate change.

Plain Language Summary Clouds play a critical role in Earth's climate, both reflecting incoming sunlight and trapping outgoing infrared radiation. Hence, even small errors in the representation of clouds in climate models can lead to uncertainty in predictions of, for example, sea ice extent. In the Arctic, liquid clouds often exist below 0°C and cloud water droplets can exist in a supercooled liquid state. In the absence of special particles that can trigger ice formation in droplets, ice-nucleating particles (INPs), supercooled water droplets can cool well below -35° C before spontaneously freezing. The presence of INPs can reduce the lifetime of a cloud and the amount of supercooled water in clouds, making them less reflective. Based on our knowledge of INPs in other remote oceans, we expected very low INP concentrations in the central Arctic. However, we found very high concentrations of biological INPs in the summertime North Pole. Furthermore, these INPs come from the seas off the coast of Russia, a region already experiencing strong climate change. It is possible that these sources may become even more important as the Arctic becomes increasingly ice-free, causing changes in Arctic clouds and further changes in climate.

1. Introduction

The Arctic climate is strongly influenced by ubiquitous low-level mixed-phase clouds (Kay & L'Ecuyer, 2013; Tjernstrom et al., 2012; Vüllers et al., 2021). The radiative effect of these clouds is influenced by the amount of ice and supercooled water they contain, which depends on an intricate balance of dynamical and microphysical processes (Morrison et al., 2012). Realistic representation of these processes is needed to correct model



Funding acquisition: Matthew E. Salter, Julia Schmale, Paul Zieger, Benjamin J. Murray

Investigation: Grace C. E. Porter, Michael P. Adams, Matthew E. Salter, Karolina Siegel, Mark D. Tarn, Julika Zinke, Benjamin J. Murray Methodology: Grace C. E. Porter, Ian M. Brooks, Caroline Leck, Matthew E. Salter, Julia Schmale, Sebastien N. F. Sikora, Paul Zieger, Benjamin J. Murray Project Administration: Matthew E. Salter, Paul Zieger, Benjamin J. Murray Resources: Grace C. E. Porter, Benjamin J. Murray

Software: Grace C. E. Porter, Luisa Ickes Supervision: Ian M. Brooks, Caroline Leck, Paul Zieger, Benjamin J. Murrav Validation: Grace C. E. Porter, Paul Zieger, Benjamin J. Murray Visualization: Grace C. E. Porter. Benjamin J. Murray Writing - original draft: Grace C. E. Porter, Benjamin J. Murray Writing - review & editing: Grace C. E. Porter, Michael P. Adams, Ian M. Brooks, Luisa Ickes, Linn Karlsson, Caroline Leck, Matthew E. Salter, Julia Schmale, Karolina Siegel, Sebastien N. F Sikora Mark D Tarn Jutta Vüllers Heini Wernli, Paul Zieger, Julika Zinke, Benjamin J. Murrav

biases in the amount of supercooled liquid in mixed-phase clouds and reduce uncertainty in feedbacks (Tan & Storelvmo, 2019).

A rare subset of the total aerosol particle population, ice-nucleating particles (INPs), can induce primary ice production in Arctic mixed-phase clouds when immersed in supercooled cloud droplets (Murray et al., 2012). In the summertime, Arctic marine atmospheric boundary layer temperatures are usually much warmer than those required for homogeneous freezing ($\leq -35^{\circ}$ C; Herbert et al., 2015), hence heterogeneous nucleation on INPs determines the primary production of ice in clouds, at least in the absence of ice precipitating from overlying clouds (Vassel et al., 2019). Numerous INP types that can induce ice nucleation over a broad range of temperatures have been identified (Hoose & Möhler, 2012; Kanji et al., 2017; Murray et al., 2012). However, the sources and ice-nucleating properties of INPs in the Arctic, especially the central Arctic (>80°N), are poorly defined.

INP measurements have been made around the periphery of the Arctic circle from locations close to, or on, land, but relatively few measurements have been made in the summertime central Arctic Ocean (see compilations in Welti et al. (2020) and Murray et al. (2021)). Recent research suggests that there are significant terrestrial sources of Arctic INPs including glacial dust from Svalbard (Tobo et al., 2019) and Iceland (Sanchez-Marroquin et al., 2020), terrestrial biological aerosol from boreal forests (Schneider et al., 2021), and even particles released from thawing permafrost (Creamean et al., 2020). There are also numerous other high latitude dust sources that have not been investigated in terms of their ice-nucleating ability (Bullard et al., 2016). Marine biogenic INPs emitted from the sea surface through bubble bursting are also thought to contribute to the INP population of the oceanic high-latitudes (Bigg, 1996; Bigg & Leck, 2001; Hartmann et al., 2020, 2021; Ickes et al., 2020; Irish et al., 2017; Wilson et al., 2015). Sea spray is thought to produce relatively low INP concentrations, but in the absence of other sources it can dominate the INP population (McCluskey, Hill, et al., 2018; Vergara-Temprado et al., 2017).

Ground level observations at several land-based sites around the Arctic throughout the year showed the highest (but variable) INP concentrations during spring, summer and autumn and the lowest concentrations in winter (Wex et al., 2019). These measurements suggest that there are both marine and terrestrial INP sources around the Arctic, but it is unclear how important these sources are for clouds over the summertime central Arctic Ocean. Based on back trajectory analysis of INP measurements in the central Arctic, Bigg (1996) suggested that there is an open ocean source of INPs active at -15° C. Later, Bigg and Leck (2001) suggested the pack ice edge and bubble bursting in local leads throughout the pack ice can serve as a source of INPs. Indeed, it has been shown that there is a reservoir of INPs in the seas around the Arctic (Creamean et al., 2019; Hartmann et al., 2021; Irish et al., 2017; Wilson et al., 2015) and INP concentrations in the central Arctic decrease during the transition from Arctic summer to autumn, possibly due to the reduced availability of ice-free marine sources (Bigg & Leck, 2001).

While it is clear that there are strong sources of INPs in the lower Arctic environment ($\leq 80^{\circ}$ N), it is not clear if these INPs are transported to the central Arctic. The prevailing view is that aerosol within the summertime high Arctic boundary layer experiences little effect from long-range transport (Kupiszewski et al., 2013), and with few sources of primary aerosol in the central Arctic Ocean, sources such as local leads may be important (Bigg & Leck, 2001). However, it has also been suggested that aerosol particles can be transported from lower latitudes into the central Arctic boundary layer either through boundary layer transport or within the free troposphere with subsequent entrainment down into the boundary layer (Igel et al., 2017; Morrison et al., 2012; Schmale et al., 2021).

The structure of the Arctic summertime boundary layer is complex (Figure 1). The boundary layer is typically several hundred meters to over a kilometer deep, but often consists of two distinct layers: the surface mixed layer and the cloud mixed layer. These two layers are each well mixed, but often separated by a decoupling layer at \sim 100–300 m that prevents efficient transport between them (Brooks et al., 2017; Shupe et al., 2013; see Figure 1). Hence, measurements at the surface are not necessarily representative of those in the cloud mixed layer. In Figure 1, we illustrate two Arctic mixed phase cloud types, one in the surface and one in the cloud mixed layer, and illustrate the three pathways we hypothesise that INP may arrive in the central Arctic in green.

In this study, we present measurements of INP concentrations close to the North Pole both in and above the surface mixed layer. The measurements were made during the Microbiology-Ocean-Cloud-Coupling in the High Arctic (MOCCHA) campaign, which took place throughout August and September 2018 on the Swedish





Figure 1. Central Arctic boundary layer structure and potential sources of INPs. The lowest part of the boundary layer (surface mixed layer) is often decoupled from the rest of the boundary layer (Brooks et al., 2017). In this paper we report INP measurements both in the surface mixed layer and in the cloud mixed layer and use these measurements to infer information about the dominant sources of INPs in the central Arctic boundary layer.

icebreaker Oden. Measurements took place while Oden was on route to the North Pole as well as when it was moored to an ice floe in the inner pack ice and drifting passively between 88 and 90°N. Samples were collected for INP analysis at both ship level (in the surface mixed layer) and using a balloon-borne sampler in the cloud mixed layer. We use backward trajectories alongside other measurements to suggest that the source of the most active INPs reaching the North Pole is outside of the pack ice, and near the Arctic coast of Russia.

2. Methods

To determine the INP concentration spectra relevant for mixed-phase clouds in the central Arctic, 48 days of sampling were conducted aboard the Swedish icebreaker Oden during Arctic summertime and the early freeze-up period (August and September). Filter samples were collected and analyzed during the journey toward the North Pole from Svalbard whilst ice breaking, and whilst moored to an ice floe.

2.1. Aerosol Sampling From the Ship and the Balloon Borne Platform

For the ship-based aerosol sampling, filters ($0.4 \,\mu$ m pore size, polycarbonate, Nuclepore Track-Etched Membrane Filters, Whatman) were used to collect aerosol by subsampling from a heated whole-air inlet at a flow rate of 9 L min⁻¹ (standard temperature and pressure). The inlet was mounted on the 4th deck of the ship, 25 m above mean sea level. This type of filter has been used previously for INP sampling and has a low background INP count and high particle recovery rates (Adams et al., 2020; O'Sullivan et al., 2018; Sanchez-Marroquin et al., 2021). In addition, these filters collect aerosol across the full atmospheric size distribution with high efficiency, despite having pores of 0.4 μ m (smaller aerosol particles are efficiently captured on the filter surface through diffusional processes; Adams et al., 2020).

Aerosol samples from a balloon-borne sampler, the selective-height aerosol research kit (SHARK; Porter et al., 2020), were collected above the surface mixed layer. All inlets were covered until sampling was started via a radio signal from the ground. Two cascade impactors were used to sample aerosol. Impactor 1 sampled at 9 L min⁻¹ (Sioutas, SKC Ltd., UK) and collected on four stages (0.25–0.5 μ m, 0.5–1.0 μ m, 1.0–2.5 μ m and 2.5–~10 μ m). For substrates on which aerosol were collected we used 25 mm diameter filters of pore size 0.05 μ m (Nuclepore track-etched membrane polycarbonate filters, Whatman, UK). In addition, we used an after filter that sits after the impactor plate to collect particles smaller than 0.25 μ m (47 mm diameter Nuclepore track-etched membrane polycarbonate filter with a pore size of 5 μ m). Impactor 2 had a flow rate of 100 L min⁻¹ (MSP Model



128, TSI, USA) and consisted of three stages $(1.0-2.5 \ \mu m, 2.5-10 \ \mu m$ and $>10 \ \mu m)$. In impactor 2 we used 75 mm diameter Nuclepore track-etched membrane polycarbonate filters of pore size 0.05 μm . INP spectra from the two overlapping stages were combined in the analysis. Also, the total INP spectrum was defined by summing across all size categories (see Porter et al. (2020)). A radiosonde (S1H2-R, Windsond, Sweden) was integrated into the instrument package to measure the temperature, pressure and relative humidity (response time of 6 s); this data could be viewed in real time from the surface. In order to choose an appropriate altitude for sampling, the radiosonde was constantly operating to provide information to the user on the ground about the SHARK altitude, boundary-layer temperature and humidity structure as the SHARK was ascending. The top of the surface mixed layer was determined from the change in gradient of temperature with altitude. In addition, sampling was paused if the relative humidity increased above 80%, and was stopped completely before the SHARK was brought back down.

2.2. INP Analysis

Filters were analyzed for INP content as soon as possible after sampling, usually within 1–12 hr of being removed from the inlet. The filter samples were not frozen before offline INP analysis, due to concerns that this may affect the INP activity. Instead, they were stored at +4°C. Performing the analysis on ship soon after sampling minimized the risk of changes in the INPs on the filter since storage at any temperature is expected to affect the activity of the samples (Beall et al., 2020). The aerosol particles on the filters were washed into either 5 or 10 mL of ultra-pure water (Millipore Alpha-Q, with a resistivity of 18 M Ω cm at 25°C) to suspend the collected aerosol particles. These particle suspensions were then pipetted to form an array of $\sim 40.1 \, \mu L$ droplets on a hydrophobic 25 mm diameter glass slide (Hampton Research) positioned on top of a cold stage, the Microliter Nucleation by Immersed Particle Instrument, μ L-NIPI (Whale et al., 2015). The μ L-NIPI is a standard INP measurement instrument that has been benchmarked alongside a range of other INP instruments during a number of intercomparison studies (DeMott et al., 2018). The cold stage cooled at a controlled rate of 1°C min-1 until all droplets had frozen, and the freezing events were recorded in order to determine the concentration of INPs with respect to the volume of air that had been sampled through the inlet. Heat sensitivity of the collected INP samples was determined by heat treatment, where subsamples of the particle suspensions in sealed 50 mL conical centrifuge tubes were immersed in a water bath at 100°C for 30 min, before being reanalyzed using the μ L-NIPI (Daily et al., 2021).

The INP concentration data presented here is shown with the contribution from the background accounted for (i.e., the background was subtracted). The background influence on the INP concentration was determined by collating the differential nucleus concentrations for water and handling blanks, and subtracting this from the sample differential nucleus concentrations. The differential concentrations were then summed to produce the cumulative INP spectra (Sanchez-Marroquin et al., 2021).

2.3. Other Measurements at Ship Level

To evaluate the concentration of dimethyl sulfide (DMS), filter samples of DMS were collected and analyzed onboard. Equivalent black carbon (eBC) concentrations were obtained from a multi-angle absorption photometer (MAAP, Model 5012, Thermo Fisher Scientific Inc.). Particle size distribution measurements were made continuously using an aerosol spectrometer (WELAS 2300HP, Palas GmbH) for particles of size 0.15–9.65 µm, and a differential mobility particle sizer (DMPS) with a custom-built medium Vienna-type differential mobility analyzer (DMA) with a mixing condensation particle counter (MCPC, Model 1720, Brechtel Manufacturing Inc.) for particles of size 10–921 nm.

An ion chromatography system (ICS-2000, Thermo Fischer Scientific, previously Dionex) was used to determine the chemical composition of the samples. Using certain standards, the concentration of chloride, nitrate, sulfate, mesylate, methane sulfonic acid, sodium, ammonium, potassium, magnesium and calcium in the sample were determined from the ion chromatograms. A synthetic sample (QC Rainwater Standard, Inorganic Ventures, USA) was used to estimate the uncertainty, which was up to 3%. More details on the method can be found in Leck and Svensson (2015).



2.4. Prevention of Ship Stack Pollution

There is contradictory evidence that combustion products in a ship's exhaust can influence INP populations, with some studies indicating that there may be a significant ice nucleating activity (Thomson et al., 2018), whereas others indicate that there is negligible activity (Welti et al., 2020). Therefore, rigorous sampling procedures were put in place in order to ensure that the INP concentrations measured were not affected by the ship stack emissions. The aerosol sampling inlets faced the ship's bow and the ship was maneuvered to face into the wind whenever the wind direction changed, which minimized the probability of sampling ship stack emissions. In addition, an autostop for the inlet pumps was operated if aerosol concentrations increased suddenly (which would be indicative of sampling the ship stack plume), halting the sampling until aerosol concentrations returned to values similar to before the pollution event. As a precaution, the direction and speed of the wind was monitored closely, and sampling was stopped when there was a chance that the wind might introduce ship stack to the sampled aerosol. Finally, sampling was stopped if any activity that could produce aerosol was planned, including the movement of the ship, ice coring, and helicopter flights (this involved the operators being on call 24 hr a day to respond to any potential contamination). Smoking of cigarettes was also only allowed in certain areas of the ship, to ensure there was no influence on aerosol sampling. We return to a discussion of ship stack emission contamination in the results section.

2.5. Backward Trajectories

In order to define the potential origin of measured INPs, backward trajectories of the air reaching an altitude of 32m at the sampling location were examined. The 10-day (only 7 days of which are used here) back trajectories were calculated using the Lagrangian analysis tool LAGRANTO (Sprenger & Wernli, 2015) with wind fields from 3-hourly operational ECMWF analyses, interpolated to a regular grid with 0.5° horizontal resolution on the 137 model levels. The trajectory data contains the hourly positions (longitude, latitude, pressure) along the trajectory. To focus on the segments of the trajectories that can potentially be affected by surface aerosol emissions, the trajectories are only included when they were within the model boundary layer. Additionally, removal of aerosol by precipitation, which may remove the signature of upwind aerosol sources via wet deposition, has been considered by removing all the trajectory points before the precipitation event (using a threshold of 0.1 mm hr⁻¹). The overall relationship with origin is unchanged by the addition of this filter, which indicates that the results were insensitive to precipitation events.

3. Results and Discussion

3.1. Ice-Nucleating Particle Concentrations Within the Surface Mixed Layer

We first present our INP concentrations derived from samples collected on the ship, which was within the surface mixed layer (Figure 2a). The concentrations of INPs measured in the surface mixed layer were highly variable, and ranged from $<6 \times 10^{-3}$ INP L⁻¹ to 2 INP L⁻¹ at -15° C. This resulted in INP activation temperatures ranging from -9 to -30° C for a concentration of 0.1 INP L⁻¹. This is clearly contrary to what we expected in this remote location based on measurements in other remote oceanic locations around the world. For example, in the Southern Ocean, INP concentrations are systematically at the low end of what we observe here (McCluskey, Hill, et al., 2018; Murray et al., 2021; Welti et al., 2020).

The vast majority of Arctic INP measurements reported in the literature were made on land, or at least with some distance from the Pole. A summary of these measurements is given Figure 2b. These measurements clearly show that there are strong sources of INP between around 65 to 80°N (Hartmann et al., 2021; Sanchez-Marroquin et al., 2020; Tobo et al., 2019; Wex et al., 2019). Our measurements demonstrate that the INP concentrations can also sporadically be very high in the pack ice near the North Pole. Previous measurements close to the Pole also reveal substantial variability in INP concentrations active at -15° C (Bigg, 1996; Bigg & Leck, 2001). Our results indicate sporadically higher concentrations than Bigg's results and also demonstrate that the concentration of INP can be higher than 0.1 L⁻¹ at temperatures up to around -10° C. We come back to the question of where these highly active INP come from later in the paper.

Overall, it is remarkable to note that the peak concentrations measured here (2 INP L^{-1} at $-15^{\circ}C$) are as high as the highest INP concentration reported in INP rich regions such as the mid-latitude terrestrial environment





Figure 2. Surface mixed layer INP concentrations throughout the campaign. (a) The number of INPs per liter of air sampled was calculated using data from offline droplet freezing experiments, conducted within hours of the samples being taken. The spectra shown in blues represent samples that were heated to close to 100° C for 30 min. Background values were subtracted from the data. Sampling times varied from 6 hr to 3 days and were taken using a heated whole air inlet on the 4th deck (25 m above mean sea level) of the Oden Icebreaker. Temperature uncertainties (not shown) for the droplet freezing experiments were estimated to be $\pm 0.4^{\circ}$ C. The format of the key is YYMMDD_hhmm; these correspond to the start time of the filter sample in the 24 hr time format; the start and end times are in Table S1 in Supporting Information S1. (b) The data from this study are presented alongside data from the literature for ground, ship and aircraft-based campaigns around the Arctic (Bigg, 1996; Bigg & Leck, 2001; Borys, 1989; Conen et al., 2016; Creamean et al., 2018, 2019; DeMott et al., 2016; Flyger & Heidam, 1978; Hartmann et al., 2020, 2021; Irish et al., 2019; Mason et al., 2016; Porter et al., 2020; Prenni et al., 2007; Rogers et al., 2001; Sanchez-Marroquin et al., 2020; Si et al., 2019; Wex et al., 2019) and a compilation derived from precipitation samples (Petters & Wright, 2015).

(O'Sullivan et al., 2018; Petters & Wright, 2015). Our measurements indicate that the INP concentration spectra within the high Arctic surface-mixed boundary layer can be extremely variable.

In order to test for the presence of proteinaceous biological ice-nucleating material, we heated the most active sample suspensions to close to 100°C (Daily et al., 2021). Heating suspensions is known to denature the proteins responsible for ice nucleation in some biological samples, whereas mineral dust is largely unaffected. The activity of these samples was always reduced, with all of the activity above -20° C being removed (Figure 2a and Figure S1 in Supporting Information S1). Hence, it appears that the most active INPs sampled were most likely of biological origin. Atmospheric INP at lower Arctic latitudes have also been found to be heat sensitive (Hartmann et al., 2021). Together, this indicates that proteinaceous biological INP are important in the Artic.

The time series in Figure 3 shows the temperature at which a concentration of 0.1 INP L^{-1} was measured $(T_{\text{(INP]= 0.1}})$, and highlights the variability of INP concentrations at the North Pole throughout August and September of 2018. The first peak in ice-nucleating activity was observed during a period in which the ship was breaking ice prior to being moored to an ice-floe (i.e., prior to 16th August). It is reasonable to question whether the very high INP concentrations observed during the ice-breaking period resulted from the ice-breaking itself. Ice-breaking involved frequent backward and forward motions of the ship, hence there is the potential for sampling ship emissions (such as ship stack emissions, detailed in the methodology) and aerosol resulting from ice-breaking and the disruption of the sea surface. See the methodology for a description of the precautions taken to avoid sampling ship stack (or other) emissions. There was a pause in ice-breaking when a clean air station was established (10th August, we sampled for 6 hr) that coincided as well with high INP concentrations. At this clean air station, the ship was moored facing into the wind, with ship-based aerosol sources downwind of the aerosol inlets. Hence, we were confident that sampling of ship pollution and ice-breaking aerosol were eliminated, increasing confidence that these high values in this period were indeed representative of the central Arctic Ocean. In addition, there was also a period of very high ice-nucleating activity a few days after the ice-floe station had been established and the ship was pointing into the wind, demonstrating that there were very active INPs that were not related to ice-breaking or ship emissions. The time series also highlights two distinct regimes: the period up to the 23rd August was







b) Comparison of ice-nucleating activity within the surface mixed layer and above



Figure 3. Time series showing the ice-nucleating activity (expressed as the temperature at which a concentration of 0.1 INP L⁻¹ ($T_{[INP]=0.1}$) was observed) throughout the campaign alongside dimethyl sulfide (DMS), aerosol surface area and equivalent black carbon (eBC). In both panels a and b, the tops of the gray bars represent $T_{[INP]=0.1}$ in the surface mixed layer (i.e., at ship level, 25 m above mean sea level), with the width of the bar representing the period over which air was sampled. The hatched gray bars are limiting values (where droplet freezing was indistinguishable from the control experiments). (a) The time series of the daily average surface area of aerosol per liter (blue), the DMS concentration (green) and the eBC concentrations (red) measured in the aerosol are shown. We also show the Pearson's r coefficient between ice-nucleating activity and each quantity in the figure legend. (b) The values of $T_{[INP]=0.1}$ measured above the surface mixed layer (using the SHARK balloonborne sampler) are shown against $T_{[INP]=0.1}$ measured at ship level (gray bars). The red triangles are the $T_{[INP]=0.1}$ for the summed INP concentrations across all size categories (comparable to the measurements at ship level), while the crosses indicate the $T_{[INP]=0.1}$ associated with each size category (circles indicate limiting values). The dates for the respective periods are: MIZ 02/08/18-03/08/18, Clean-air station 10/08/18-11/08/18, Ice-breaking 03/08/18-16/08/18, Ice floe station 16/08/18-15/09/18, Ice-breaking 15/09/18-19/09/18, MIZ 19/09/18.

characterized by variable but often very high INP concentrations, whereas after this date INP concentrations were typically much lower. We examine the back trajectories associated with these different regimes later in the paper.

We also show the ice-nucleating activity in the form of ice-active sites per unit surface area (n_s) of aerosol particles in Figure 4. To calculate n_s we used the total aerosol surface area derived from the online DMPS and WELAS instruments sampling from the same whole air inlet that we used to collect the filter samples. n_s provides a means of comparing the activity of aerosol on a per unit surface area basis and we compare our results to parameterisations for marine aerosol (McCluskey, Ovadnevaite et al., 2018), desert dust (Ullrich et al., 2017), and high latitude dust from Iceland (Sanchez-Marroquin et al., 2020). It is striking that the activity of our samples in the central Arctic are often much higher than aerosol in the north Atlantic or from over the Southern Ocean (which have a similar activity to that reported for the north Atlantic (McCluskey, Hill, et al., 2018)). It is perhaps even more striking that a subset of the samples from the central Arctic are more active than mineral dust aerosol from Iceland





Figure 4. The ice-active site density (n_s) for aerosol sampled during the cruise (see Figure 2 for key to colors) compared to n_s parameterisations for desert dust (Ullrich et al., 2017), Icelandic dust (Sanchez-Marroquin et al., 2020) and for aerosol from the North Atlantic (McCluskey, Ovadnevaite, et al., 2018), which is also consistent with values for aerosol over the Southern Ocean (McCluskey, Hill, et al., 2018).

or low latitude deserts. This shows the aerosol at this location can be much more ice-active than aerosol in other remote marine environments.

3.2. Ice-Nucleating Particle Concentrations Above the Surface Mixed Layer

Eight flights with a balloon-borne size-resolved aerosol sampler were conducted while the Oden was at the ice floe station (see SI for flight details). The sampler (SHARK), was lofted to a defined height using a tethered balloon operated from the sea ice, and aerosol samples were collected into multiple size bins (see methods and Porter et al. (2020)). During these flights, we used a live link to the on-board temperature and humidity measurements to ensure that we sampled above the surface mixed layer and thus in air decoupled from the surface, but within the boundary layer (i.e., in the cloud mixed layer). Flights occurred at 390–600 m altitude, each sampling for 3–6 hr. Sampling was paused if the RH increased above 80% to avoid sampling biases associated with hygroscopically swollen aerosol and sampling in cloud.

Given the surface mixed layer is often decoupled from the rest of the boundary layer, these measurements in principle allow us to compare INP concentrations within and above the surface mixed layer. The values of $T_{[INP]=0.1}$ are shown in Figure 2b, and the INP spectra are shown in Figure S2 in Supporting Information S1 summed across all size bins. It should be borne in mind that, for practical reasons, the sampling durations on the ship and on the SHARK were not the same. However, it is still possible to draw conclusions from this comparison. There is evidence that there are substantial differences

between the INP concentrations in the surface mixed layer compared to above it. For example, on the 5th and 8th September the $T_{[INP]=0.1}$ (summed across all sizes) was around -18 to -19° C above the surface mixed layer, whereas it was below -26° C within it. On the 20th August, $T_{[INP]=0.1}$ was around -23° C above the surface mixed layer, but -14° C within it. On all three days, scanning microwave radiometer and six hourly radiosonde temperature profiles confirmed that the surface mixed layer was decoupled from the rest of the boundary layer (Table S2 in Supporting Information S1). In contrast, on 13th September, the boundary layer was mainly coupled and the INP concentrations within and above the surface mixed layer and above were similar. However, on two other days (23rd August and 10th September) the activity in the surface mixed layer and above were similar even though it was decoupled. Overall, out of the eight SHARK samples collected above the surface mixed layer, three was one SHARK sample that had much lower ice-nucleating activity than that in the surface mixed layer, three samples with higher activity, three with similar activity, and one that was ambiguous (due to both samples being close to the baseline of detection). This is consistent with the air at the surface sometimes being coupled to the cloud mixed layer, allowing transport of aerosol throughout the boundary layer, but at other times the measurements at the surface are not representative of those above the surface mixed layer.

The size-resolved INP activity ($T_{[INP]=0.1}$) is also shown in Figure 3b. In many locations around the world, supermicron aerosol dominate the INP population (Creamean et al., 2018; Gong et al., 2020; Mason et al., 2016; Porter et al., 2020; Reicher et al., 2018). However, contrary to what might be expected, the smallest size ranges of <0.25 µm contributed the most INPs on five out of the eight flights, with the 2.5–10 µm and 0.5–1 µm bins both contributing the most on one flight each. Inspection of the corresponding INP spectra associated with each bin (Figure S2 in Supporting Information S1) revealed that the particles <0.25 µm made a pronounced contribution to the INP population on the 23rd August and the 8th and 9th September. The other flights produced data mainly in the baseline for all sizes.

The coarse mode (>2.5 μ m diameter) has a relatively short lifetime in the Arctic boundary layer, being removed effectively by wet scavenging processes (Leck & Svensson, 2015). Hence, it is perhaps not so surprising that the fine mode aerosol (<0.25 μ m) appears to be so important in this region for the INP population. While INPs are typically thought of as being the larger particles in a size distribution (Mason et al., 2016; Porter et al., 2020), there are INPs that fall into the <0.25 μ m size range that are also very active. For example, film droplet aerosol resulting from wave breaking are produced in a range of sizes centered around 100–200 nm and are often rich



in organic material (Bigg & Leck, 2008; Leck & Bigg, 2005; O'Dowd et al., 2004; Orellana et al., 2011) that is thought to include small ice-nucleating entities (Ickes et al., 2020; Schnell & Vali, 1975; Wilson et al., 2015). Ice-nucleating macromolecules from terrestrial biological sources internally mixed with other aerosol particles might also fall into this size range (O'Sullivan et al., 2016; O'Sullivan et al., 2015; Pummer et al., 2015) and it has been proposed that fungal material, some of which is known to act as an INP (O'Sullivan et al., 2015), can fragment to form nanoparticles (Lawler et al., 2020).

3.3. Correlation Between INP Concentrations and Dimethyl Sulfide, Equivalent Black Carbon and Aerosol Surface Area

To investigate possible sources of the INPs we detected over the central Arctic Ocean, we correlated the ice-nucleating activity of the aerosol with: (a) dimethyl sulfide (DMS), a product of marine biological activity, particularly in the marginal ice zone (MIZ, the transitional zone between open sea and dense pack ice; Leck & Persson, 1996); (b) eBC, based on aerosol absorption at 637 nm; and (c) aerosol surface area, derived from size distribution measurements. We present the time series for aerosol particle surface area, DMS and eBC concentrations, as well as the Pearson's r coefficient between ice-nucleating activity and each quantity in Figure 3a.

DMS is found in the marine atmosphere, originating from the metabolites of some marine algae (Leck & Persson, 1996; Lohmann & Leck, 2005). Hence, the presence of DMS indicates that an air mass has origins in a location rich in biological activity, which may also be expected to correlate with marine biological INP sources. DMS is thought to be relatively short-lived in the atmosphere, with a lifetime on the order of 1–3 days (Kerminen & Leck, 2001; Khan et al., 2016). Therefore, it is a useful indicator for the interaction of air masses with the MIZ at the outer edge of the pack ice region, and possibly the open leads or melt ponds within the pack ice if they were producing DMS at that time.

The concentration of DMS during the cruise was highest during the outbound 24 hr MIZ station (2nd-3rd August), where the ship was close to open water, but was variable whilst in the pack ice (Figure 3), and remained relatively low during the inbound MIZ station (19th September). The data in Figure 3 clearly shows that there is no obvious correlation between DMS and INP activity (r = 0.15) suggesting that MIZ marine biogenic sources exerted little influence on the measured INP concentrations. This does not rule out marine sources of INP further afield.

Equivalent black carbon (eBC) is a quantity derived from aerosol absorption and is the equivalent black carbon mass concentration needed to produce the observed absorption. Other aerosol types such as dust, brown carbon or other organic aerosol might also produce absorption, thus potentially contaminating the small signal we observed. However, absorption at 637 nm by BC is much stronger than for other materials, hence the signal is most likely dominated by BC. BC is produced through a range of combustion processes, including biomass burning, wildfires and fossil fuel combustion, which all occur remote from the central Arctic. Other potential contributors to the absorption signal, such as dust or brown carbon are also remote from the central Arctic. Rigorous procedures were in place to ensure that BC (and other aerosol) from the ship stack did not affect measurements (see methods for details). Therefore, eBC is used here as an indicator of long-range transport. The literature indicates that BC is a relatively ineffective ice nucleator under mixed-phase cloud conditions (Adams et al., 2020; Chen et al., 2018; Schill et al., 2020; Vergara-Temprado, Holden et al., 2018; Welti et al., 2020), hence we would not necessarily interpret a positive correlation as an indication of ice nucleation by BC. However, combustion processes are thought to be a source of ice-nucleating aerosol, even if BC itself is not an effective INP (Barry et al., 2021; Jahn et al., 2020; Umo et al., 2015; Wagner et al., 2018). Thus, a correlation between BC and INP concentrations would indicate that aerosol particles transported along with BC from outside the area of the central Arctic Ocean nucleate ice. Wildfires around the Arctic are a potential source of BC, and we note that during the cruise there were persistent Siberian wildfires, as can be seen using the NASA Worldview satellite imagery tool (e.g., clear skies on the 14th August reveal widespread fires; https://worldview.earthdata.nasa.gov/). There is also industry, shipping and mining along the Arctic coast of Russia, and the role of gas flaring is thought to be a particularly important source of BC (Stohl et al., 2013). The overall correlation between eBC and ice-nucleating activity (expressed as $T_{\text{[INP]}=0.1}$) is 0.65. There are periods when the eBC concentration and $T_{\text{[INP]}=0.1}$ track one another, but there are other periods where they vary independently. For example, between the 6th and the 19th August eBC and $T_{\text{IINP}=0.1}$ are highly correlated. However, on the 20th August the eBC decreases dramatically, but the INP remain elevated. Conversely, later in the campaign, the eBC values are some of the highest we measured, but the ice nucleating activity remains very low. Hence, we concluded that the eBC indicates long range transport, but this transport only sometimes coincides with remote sources of INP.

The surface area concentration of the bulk aerosol follows a similar trend to the eBC concentrations, but with a slightly weaker correlation with the $T_{[INP]=0.1}$ (r = 0.52). Similar to the eBC, the surface area and $T_{[INP]=0.1}$ 6th and the 19th August correlate strongly, but on the 20th August the aerosol surface area decreases by around an order of magnitude without a corresponding decrease in $T_{[INP]=0.1}$. This indicates that the variability in INP concentrations at the North Pole is not simply driven by aerosol surface area, rather that some specific component(s) of the aerosol population are ice-active and these particles are most likely associated with specific sources at latitudes further south than the MIZ.

3.4. Trajectory Analysis of Aerosol Collected in the Central Arctic

Backward trajectories from the sampling location near the surface are presented in Figure 5a. We only show points that are within in the boundary layer and for a maximum of 7 days. The figure shows a clear relationship between the origin of the aerosol, considered here as the boundary layer points along the trajectories, and the measured INP concentrations. The origin of the air with the most active INPs is around the Russian Arctic coast including the Barents, Kara and Laptev Seas. Out of the 30 filter runs, those filters with the highest INP activity (the top 20% of filters; $-9^{\circ}C \ge T_{[INP]= 0.1} > -13.5^{\circ}C$) sampled air masses originating over the Barents and Kara Seas. The next seven highest (23%; $-13.5^{\circ}C \ge T_{[INP]= 0.1} > -22^{\circ}C$) filters, in terms of INP activity, sampled air originating from over the Laptev and East Siberian Seas. The next six filters with lower INP activity (20% of filters; $-22^{\circ}C \ge T_{[INP]= 0.1} > -25^{\circ}C$) sampled air that originated off the eastern coast of Greenland from over both the pack ice and open ocean. The 11 filters with the lowest INP activity (bottom 37% of filters; $-25^{\circ}C \ge T_{[INP]= 0.1} \ge -30^{\circ}C$) all sampled air which mostly originated from the pack ice adjacent to North America (also see Figure S3 in Supporting Information S1).

It is striking that the trajectories with the lowest INP concentrations spent most of the preceding seven days over the pack ice and to some extent over the MIZ. These results indicate that, during this campaign, open leads, sea ice and the MIZ were at most weak sources of INPs. This conclusion contrasts with previous suggestions (Bigg & Leck, 2001; Hartmann et al., 2020) of a local source of biogenic particles and INP from open leads.

The highest ice-nucleating activities from sampled aerosol originating along the Russian coast were also correlated with high wind speeds along the trajectories, that is, in the region of the aerosol origin (Figure 5b). This, together with the heat tests and INP size information, point to a wind-driven marine biological source of INPs associated with organic-rich film droplet sea spray aerosol. There were trajectories with high wind speeds over the North American continent, the pack ice and the coast of Greenland, but the ice-nucleating activity for these was not greatly enhanced. Hence, our results are consistent with a strong source of highly active INPs in the coastal marine waters of northern Russia that were aerosolized during windy conditions. Marine waters elsewhere in the world are thought to produce aerosol with relatively low ice nucleating activities (Vergara-Temprado et al., 2017; see Figure 4); however, our results suggest that the shallow seas off the Russian coast might be relatively strong sources of highly active INPs. Composition analysis of the aerosol during the peak ice nucleation activity on the 11th August and the 19th August is consistent with a marine source for many of these aerosol particles (samples were rich in Na, Cl and sulfate; see Table S3 in Supporting Information S1). Hence, the question is why the aerosol from near the coast of Russia is so much more active than aerosol derived from other marine locations such as the North Atlantic and the Southern Ocean (McCluskey, Hill, et al., 2018; McCluskey, Ovadnevaite, 2018).

A major difference to the North Atlantic and Southern Ocean is that the shallow seas off the coast of northern Russia are strongly influenced by riverine input from Russia that is rich in organic material, silt and nutrients (Ahmed et al., 2020; Juhls et al., 2019). In fact, much of the dissolved organic matter in the Arctic Ocean is derived from river input (Juhls et al., 2019) and the discharge of these rivers (and the amount of dissolved organic carbon flowing into the sea via rivers) is increasing (Ahmed et al., 2020; Juhls et al., 2020). It has been shown that melting permafrost, which is known to enter river water (Juhls et al., 2020), harbors copious quantities of warm temperature INPs (Creamean et al., 2020). Also, it has been shown that river water elsewhere in the world can contain highly active INPs (Knackstedt et al., 2018; Larsen et al., 2017). Hence, it is possible that the highest INP concentrations we detected at the North Pole were derived from marine waters rich in terrestrially derived ancient biological INPs. Alternatively, marine biology fertilised by the nutrient rich waters on the continental shelf may





Figure 5. Backward trajectories over 7 days, starting at the ship location, for the INP samples taken throughout the campaign. Trajectories were launched every hour during the sampling period, and each point represents an hour in time along the back trajectory. The starting height for the trajectories was 32 m above mean sea level. Any points along the trajectories which were above the model boundary layer and any points along the trajectory that preceded precipitation events (>0.1 mm hr⁻¹) were removed. Hence, any potential sources of INP in the boundary layer are neglected if they occurred prior to a precipitation event (we assume precipitation removes INPs). (a) The color of the trajectories represents the temperature at which 0.1 INP L⁻¹ was measured for that sampling period. (b) The color of the trajectories represents the wind speed for each point along the trajectory. The sea ice extent is from the NASA National Snow and Ice Data Center (Maslanik & Stroeve, 1999).





Time over Ice (> 70% ice)

Figure 6. The time that air masses spent over water, ice and land derived using back trajectory analysis. The color represents the temperature at which an INP concentration of $0.1 \text{ INP } \text{L}^{-1}$ was reached. The black dots correspond to individual trajectories. The same filters as in Figure 5 were applied.

produce more INPs than are present in remote marine locations. A measurement campaign to quantify the INP content of the waters off the coast of Russia is clearly required.

Some of the back trajectories that had the highest INP concentrations passed over islands in the Barents and Kara Seas, including Svalbard, Franz Josef Land, Novaya Zemlya and Severnaya Zemlya. Many of these locations have been identified as poorly defined dust sources (Bullard et al., 2016; Meinander et al., 2021) and dust from Svalbard has been shown to contain biological ice-nucleating materials (Tobo et al., 2019). However, in a further analysis of the back trajectory data (Figure 6), we find that there was little to no correlation with time spent over land, whereas the ice-nucleating activity increased with the time the air parcels spent over open ocean. This implies that the sources of INP were associated with the marine environment. Having said this, we cannot rule out relatively small island point sources being important for INPs.

Overall, the evidence indicates that there is a strong source of biogenic INPs in the Barents, Kara and Laptev Seas off the Russian coast that can be sporadically transported to the central Arctic Ocean. There was high wind along the trajectories off the Russian coast which would be consistent with both the production of INPs in sea spray from these organic-rich seas or dust combined with terrestrial biogenic material from the various islands in this region. We also note that a recent study found that very active INP were produced in the Chukchi sea (near Alaska) under high wave conditions (Inoue et al., 2021). Furthermore, we note that these potential sources are sensitive to a changing climate, with river discharge, permafrost melt and organic matter input into the ocean from the major Russian rivers increasing in a warmer world (Jahn et al., 2020). In addition, the removal of ice from the Arctic could also expose marine and terrestrial sources of ice-nucleating material around the Arctic Ocean, where it can then be aerosolized by the action of wind (Schmale et al., 2021).

3.5. Implications for Ice Production in Boundary Layer Central Arctic Mixed-Phase Clouds

In this section, we assess whether the measured INP concentrations are high enough to initiate a transition from liquid-dominated clouds to ice-dominated clouds. Model simulations indicate that on the order of one ice crystal per liter of air is required to remove the bulk of liquid water from an Arctic cloud, whereas lower concentrations still reduce the liquid water path (Stevens et al., 2018; Vergara-Temprado, Miltenberger et al., 2018). Hence, we use our INP measurements, both at ship level in the surface mixed layer and those from the SHARK in the cloud





Figure 7. Time series showing [INP]_{ambient} for both the measurements taken at ship height (within the surface mixed layer) and using the SHARK (within the cloud mixed layer). The temperature at the top of the mixed layers is shown alongside these measurements. When an INP spectrum did not extend to the atmospheric temperature (i.e., where the highest temperature at which an INP concentration was reported was below the atmospheric temperature), the [INP] value associated with the highest temperature is provided as an upper limit to [INP]_{ambient} (open symbols).

mixed layer, to estimate the concentration of INPs that become active at the ambient temperature of the atmosphere ([INP]_{ambient}).

The quantity $[INP]_{ambient}$ combines the atmospheric temperature profiles from radiosonde measurements (six hourly) with the corresponding INP spectra as a rough indicator of primary ice crystal production. This is a crude analysis – a full cloud model would be required to represent ice crystal formation and sedimentation as well as INP recycling and latent heat release, and to predict ice crystal concentration given an INP spectrum – but it does give an indication of what the measured INP spectra might mean for primary ice production in clouds. The highest $[INP]_{ambient}$ will be at the coldest points, that is, the top of the surface mixed layer and the top of the boundary layer (top of the cloud mixed layer), hence this is where we focus this analysis. We assume that ship level measurements are representative of the $[INP]_{ambient}$ at the top of the surface mixed layer and those from SHARK are representative of the top of the boundary layer, the assumption being that these respective layers are individually well-mixed.

The temperature minima within the main boundary layer were determined from routine radiosonde profiles with Väisälä RS92 radiosondes, which were made every six hours throughout the entire cruise (Vüllers et al., 2021). The calculated [INP]_{ambient} for the duration of the cruise are shown in Figure 7, where the minimum temperature at the top of the cloud mixed layer (main boundary layer) and the top of the surface mixed layer is indicated for each filter period. In many cases, the temperature of the atmosphere was higher than the temperatures at which we report the INP concentrations, hence we are only able to report upper limits to [INP]_{ambient} in these cases.

Generally, $[INP]_{ambient}$ was typically below about 0.1 INP L⁻¹ at the top of both the surface mixed layer and the top of the cloud mixed layer (and since many of the values in Figure 7 were upper limits, the $[INP]_{ambient}$ might be much smaller than 0.1 L⁻¹). The periods of high INP concentration before the 23rd August coincide with periods of higher ambient inversion temperatures, whereas later in the campaign the opposite is the case, which results in a relatively invariant $[INP]_{ambient}$. Whether this is a coincidence, or if INP concentrations are correlated with ambient temperature, is unclear from this limited data set. However, there may be a physical mechanism behind this apparent correlation. Transport of air to the North Pole from sources further afield will likely result in multiple cycles of cloud formation and dissipation in any one air mass, hence INP active at above the ambient temperature of those clouds will likely activate and be removed via precipitation (bearing in mind that the majority of INPs relevant for mixed-phase clouds only activate to ice in the presence of water droplets (Murray et al., 2012)). While transport through the boundary layer likely removes INP active above the lowest temperatures experienced by an



air parcel, further cooling subsequent to the measurement time will lead to primary ice production. Hence, the air masses sampled before the 23rd August with high INP concentrations have a great potential for primary ice production if or when these air masses become colder downwind of the sampling location. In the absence of local sources, the INP spectrum over the central Arctic Ocean must therefore be determined by a combination of the characteristics of the upwind sources and the cloud temperature that these air parcels experience during transport.

The relatively low $[INP]_{ambient}$ values suggest that clouds (assuming that they are not influenced by seeding from above) should be mixed-phase, that is, contain a substantial proportion of liquid water with some ice crystals. Observations of the phase of clouds during this campaign are discussed in Vüllers et al. (2021). Overall, the fraction of single layer clouds (where seeding from above is unlikely) throughout the troposphere that were mixed-phase was relatively constant throughout the campaign: mixed phase frequency ~20%-30% (of the campaign period), ice cloud frequency ~10%-30% and liquid clouds were generally infrequent (see Figure 15 in Vüllers et al. (2021)). The mixed-phase frequency was relatively constant despite the strong decrease in temperature during the campaign. Similarly, in the bottom few kilometers the frequency of occurrence of mixed-phase clouds were often between 25% and 50%, with ice clouds about half as frequent and liquid only clouds much less frequent throughout the campaign. Overall, the observations of a lack of clear trend in cloud phase for single-layer clouds reported by Vüllers et al. (2021) are consistent with our relatively invariant [INP]_{ambient} measurements.

Clouds were multi-layered around 50% of the time, with many situations identified where seeding of ice from higher, colder clouds into lower clouds might occur. Clouds were regularly observed up to around 8 or 9 km, where temperatures were low enough for homogeneous freezing (Herbert et al., 2015), and frequently occurred in the mid-troposphere where heterogeneous nucleation on INPs was most likely important. These higher clouds were often in the free troposphere where our boundary layer INP measurements are not necessarily relevant. Indeed, aerosol and INPs in the free troposphere may have different sources to those in the boundary layer. In order to obtain a more complete picture of primary ice production in clouds in the central Arctic, INP measurements in the free troposphere in this region would be needed and should be a target of future campaigns.

4. Summary and Conclusions

Arctic mixed-phase and supercooled clouds play a crucial role in Arctic climate, but the processes that dictate their characteristics are poorly understood. Here, we show that INP concentrations at 88–90°N are extremely variable, and throughout the MOCCHA campaign between the 1st of August 2018 and the 18th of September 2018 the temperature at which 0.1 INP L^{-1} was reached varied between $-9^{\circ}C$ and $-30^{\circ}C$. The highest 20% of observed INP activity is related to air masses originating in the ice-free ocean environment off the Russian coast, while the lowest 37% of observations related to air masses which originated and circled over the pack ice north of Canada for most of the 7-day back trajectory. Trajectories of air with intermediate INP activity also originated over the ice-free ocean. These results indicate a strong dependence of the measured INP concentration on the origin of the air with pack ice, open leads, and the MIZ apparently being weak sources of INP, whereas ice-free oceans, especially those near the Russian coast when wind speeds were high, were a significant source.

The heat sensitivity of the most active INPs indicates the INP to be of biogenic proteinaceous origin. This, together with the trajectory analysis, indicates that there are strong biogenic sources of INP in the shallow seas over the Russian continental shelf. The ice-nucleating activity of the aerosol at the North Pole derived from off the coast of Russia is much greater than that for sea spray aerosol from other oceans (including the Southern Ocean (McCluskey, Hill, et al., 2018) or the North Atlantic (McCluskey, Ovadnevaite, et al., 2018)). This may indicate the marine waters off Russia are very rich in ice-nucleating material, perhaps related to the substantial riverine input, or alternatively the islands in this region may be sources of biogenic INPs. More work is needed to define what the key sources are along the Russian coast and to see if similar sources exist elsewhere around the Arctic and Antarctic.

By making measurements of INP spectra both above and within the surface mixed layer of decoupled boundary layers, we found that surface measurements were often not representative of the INPs in the cloud mixed layer. Hence, measurements at altitude, within the cloud mixed layer, are necessary in order to define primary ice production in Arctic mixed-phase clouds. In addition, our measurements allowed us to estimate the INP concentration active at the temperature of the top of the surface mixed layer and also at the top of the boundary layer. This revealed that, despite massive variability in INP spectra, the INP concentration at ambient temperature was



typically less than $0.1 L^{-1}$, which is consistent with remote sensing observations that indicate the persistence of mixed-phase clouds (in the absence of seeding of ice from above). We also recommend future studies focus on INP measurements throughout the free troposphere where primary ice production may lead to seeding of ice in lower level clouds.

Overall, it is striking that INP concentrations at the summertime North Pole vary from some of the lowest measured anywhere in the world, to as high as the highest INP concentrations in terrestrial locations rich in biological INPs such as in the UK (O'Sullivan et al., 2018). Since these INPs are transported from the seas off the Russian coast, they may be sensitive to changes in climate. In particular, reduced sea ice, land ice, and permafrost may open up more INP sources for more of the year around the Arctic, which may increase the future strength (and may already have done so) of the sources of INPs that are important for mixed-phase clouds in the central Arctic. More work needs to be undertaken to understand how climate change may affect INP sources around the periphery of the Arctic and how this may influence Arctic clouds and feedback on Arctic climate.

Conflict of Interest

The authors declare no conflicts of interest relevant to this study.

Data Availability Statement

The ice-nucleating particle data associated with this paper is available from University of Leeds data repository at (https://doi.org/10.5518/1093; Porter et al., 2022). The fine mode aerosol size distribution data is available from the Bolin database at https://doi.org/10.17043/oden-ao-2018-aerosol-dmps-1 (Karlsson & Zieger, 2020), the coarse mode aerosol size distribution data is available at https://doi.org/10.17043/oden-ao-2018-aerosol-coarse-1 (Zieger & Karlsson, 2022a) and the black carbon data is available at https://doi.org/10.17043/oden-ao-2018-aerosol-coarse-1 (Zieger & Karlsson, 2022a). In addition, there is an overview of all available data from the Arctic Ocean expedition, 2018 (Leck et al., 2018).

Acknowledgments

This research was part of the Arctic Ocean (AO) 2018 expedition. The Swedish Polar Research Secretariat (SPRS) provided access to the icebreaker (I/B) Oden and logistical support in collaboration with the U.S. National Science Foundation. We are grateful to the Co-Chief Scientist Patricia Matrai for planning and coordination of AO2018 (along with coauthor Leck) as well as to the SPRS logistical staff and to I/B Oden's Captain Mattias Peterson and his crew for expert field support. We are grateful for funding from the European Research Council 648661 MarineIce (BJM), Natural Environment Research Council NE/R009686/1 (IMB and BJM), NE/T00648X/1 (BJM), Swiss National Science Foundation Grant No. 200021 169090 (JS), Swiss National Science Foundation Early Postdoc. Mobility (LI), Swiss Polar Institute (JS), Knut-and-Alice-Wallenberg Foundation within the ACAS project (Arctic Climate Across Scales) project no. 2016.0024 (PZ), Bolin Centre for Climate Research, RA2 (CL.PZ, MES), Swedish Research Council project nos. 2016-03518 and 824-2013-222 (CL), 2018-05045 (PZ), 2015-05318 (LI) and 2016-05100 (MES), the Ymer-80 foundation (LI) and The Ingvar Kamprad Chair, sponsored by Ferring Pharmaceuticals (JS).

References

- Adams, M. P., Tarn, M. D., Sanchez-Marroquin, A., Porter, G. C. E., O'Sullivan, D., Harrison, A. D., et al. (2020). A major combustion aerosol event had a negligible impact on the atmospheric ice-nucleating particle population. *Journal of Geophysical Research*, 125(22), e2020JD032938. https://doi.org/10.1029/2020JD032938
- Ahmed, R., Prowse, T., Dibike, Y., Bonsal, B., & O'Neil, H. (2020). Recent trends in freshwater influx to the Arctic Ocean from four major. *Arctic-Draining Rivers*, *12*(4), 1189. https://doi.org/10.3390/w12041189
- Barry, K. R., Hill, T. C. J., Levin, E. J. T., Twohy, C. H., Moore, K. A., Weller, Z. D., et al. (2021). Observations of ice nucleating particles in the free troposphere from Western US wildfires. *Journal of Geophysical Research*, 126(3), e2020JD033752. https://doi.org/10.1029/2020JD033752
- Beall, C. M., Lucero, D., Hill, T. C., DeMott, P. J., Stokes, M. D., & Prather, K. A. (2020). Best practices for precipitation sample storage for offline studies of ice nucleation in marine and coastal environments. *Atmospheric Measurement Techniques*, 13(12), 6473–6486. https://doi. org/10.5194/amt-13-6473-2020
- Bigg, E. K. (1996). Ice forming nuclei in the high Arctic. Tellus B: Chemical and Physical Meteorology, 48(2), 223-233. https://doi.org/10.1034/j.1600-0889.1996.t01-1-00007.x
- Bigg, E. K., & Leck, C. (2001). Cloud-active particles over the central Arctic Ocean. Journal of Geophysical Research, 106(D23), 32155–32166. https://doi.org/10.1029/1999jd901152
- Bigg, E. K., & Leck, C. (2008). The composition of fragments of bubbles bursting at the ocean surface. *Journal of Geophysical Research*, 113, D11209. https://doi.org/10.1029/2007jd009078
- Borys, R. D. (1989). Studies of ice nucleation by Arctic aerosol on AGASP-II. Journal of Atmospheric Chemistry, 9(1), 169–185. https://doi.org/10.1007/BF00052831
- Brooks, I. M., Tjernström, M., Persson, P. O. G., Shupe, M. D., Atkinson, R. A., Canut, G., et al. (2017). The turbulent structure of the arctic summer boundary layer during the arctic summer cloud-ocean study. *Journal of Geophysical Research*, 122(18), 9685–9704. https://doi. org/10.1002/2017JD027234
- Bullard, J. E., Baddock, M., Bradwell, T., Crusius, J., Darlington, E., Gaiero, D., et al. (2016). High-latitude dust in the Earth system. *Review of Geophysics*, 54(2), 447–485. https://doi.org/10.1002/2016RG000518
- Chen, J., Wu, Z., Augustin-Bauditz, S., Grawe, S., Hartmann, M., Pei, X., et al. (2018). Ice-nucleating particle concentrations unaffected by urban air pollution in Beijing, China. *Atmospheric Chemistry and Physics*, *18*(5), 3523–3539. https://doi.org/10.5194/acp-18-3523-2018
- Conen, F., Stopelli, E., & Zimmermann, L. (2016). Clues that decaying leaves enrich Arctic air with ice nucleating particles. Atmospheric Environment, 129, 91–94. https://doi.org/10.1016/j.atmosenv.2016.01.027
- Creamean, J. M., Cross, J. N., Pickart, R., McRaven, L., Lin, P., Pacini, A., et al. (2019). Ice nucleating particles carried from below a phytoplankton bloom to the arctic atmosphere. *Geophysical Research Letters*, 46(14), 8572–8581. https://doi.org/10.1029/2019GL083039
- Creamean, J. M., Hill, T. C. J., DeMott, P. J., Uetake, J., Kreidenweis, S., & Douglas, T. A. (2020). Thawing permafrost: An overlooked source of seeds for arctic cloud formation. *Environmental Research Letters*, 15(8), 084022. https://doi.org/10.1088/1748-9326/ab87d3

Creamean, J. M., Kirpes, R. M., Pratt, K. A., Spada, N. J., Maahn, M., de Boer, G., et al. (2018). Marine and terrestrial influences on ice nucleating particles during continuous springtime measurements in an Arctic oilfield location. *Atmospheric Chemistry and Physics*, 18(24), 18023–18042. https://doi.org/10.5194/acp-18-18023-2018

- DeMott, P. J., Hill, T. C. J., McCluskey, C. S., Prather, K. A., Collins, D. B., Sullivan, R. C., et al. (2016). Sea spray aerosol as a unique source of ice nucleating particles. *Proceedings of the National Academy of Sciences of the United States of America*, 113(21), 5797–5803. https://doi. org/10.1073/pnas.1514034112
- DeMott, P. J., Möhler, O., Cziczo, D. J., Hiranuma, N., Petters, M. D., Petters, S. S., et al. (2018). The fifth international workshop on ice nucleation phase 2 (FIN-02): Laboratory intercomparison of ice nucleation measurements. *Atmospheric Measurement Techniques*, 11(11), 6231–6257. https://doi.org/10.5194/amt-11-6231-2018
- Flyger, H., & Heidam, N. Z. (1978). Ground level measurements of the summer tropospheric aerosol in Northern Greenland. Journal of Aerosol Science, 9(2), 157–168. https://doi.org/10.1016/0021-8502(78)90075-7
- Gong, X., Wex, H., Voigtländer, J., Fomba, K. W., Weinhold, K., vanPinxteren, M., et al. (2020). Characterization of aerosol particles at Cabo Verde close to sea level and at the cloud level – Part 1: Particle number size distribution, cloud condensation nuclei and their origins. Atmospheric Chemistry and Physics, 20(3), 1431–1449. https://doi.org/10.5194/acp-20-1431-2020
- Hartmann, M., Adachi, K., Eppers, O., Haas, C., Herber, A., Holzinger, R., et al. (2020). Wintertime airborne measurements of ice nucleating particles in the high arctic: A hint to a marine, biogenic source for ice nucleating particles. *Geophysical Research Letters*, 47(13), e2020GL087770. https://doi.org/10.1029/2020GL087770
- Hartmann, M., Gong, X., Kecorius, S., vanPinxteren, M., Vogl, T., Welti, A., et al. (2021). Terrestrial or marine Indications towards the origin of ice-nucleating particles during melt season in the European Arctic up to 83.7°N. Atmospheric Chemistry and Physics, 21(15), 11613–11636. https://doi.org/10.5194/acp-21-11613-2021
- Herbert, R. J., Murray, B. J., Dobbie, S. J., & Koop, T. (2015). Sensitivity of liquid clouds to homogenous freezing parameterizations. *Geophysical Research Letters*, 42(5), 1599–1605. https://doi.org/10.1002/2014GL062729
- Hoose, C., & Möhler, O. (2012). Heterogeneous ice nucleation on atmospheric aerosols: A review of results from laboratory experiments. Atmospheric Chemistry and Physics, 12(20), 9817–9854. https://doi.org/10.5194/acp-12-9817-2012
- Ickes, L., Porter, G. C. E., Wagner, R., Adams, M. P., Bierbauer, S., Bertram, A. K., et al. (2020). The ice-nucleating activity of Arctic sea surface microlayer samples and marine algal cultures. *Atmospheric Chemistry and Physics*, 20(18), 11089–11117. https://doi.org/10.5194/ acp-20-11089-2020
- Igel, A. L., Ekman, A. M. L., Leck, C., Tjernström, M., Savre, J., & Sedlar, J. (2017). The free troposphere as a potential source of arctic boundary layer aerosol particles. *Geophysical Research Letters*, 44(13), 7053–7060. https://doi.org/10.1002/2017GL073808
- Inoue, J., Tobo, Y., Taketani, F., & Sato, K. (2021). Oceanic supply of ice-nucleating particles and its effect on ice cloud formation: A case study in the Arctic Ocean during a cold-air outbreak in early winter. *Geophysical Research Letters*, 48(16), e2021GL094646. https://doi. org/10.1029/2021GL094646
- Irish, V. E., Elizondo, P., Chen, J., Chou, C., Charette, J., Lizotte, M., et al. (2017). Ice-nucleating particles in Canadian Arctic sea-surface microlayer and bulk seawater. Atmospheric Chemistry and Physics, 17(17), 10583–10595. https://doi.org/10.5194/acp-17-10583-2017
- Irish, V. E., Hanna, S. J., Willis, M. D., China, S., Thomas, J. L., Wentzell, J. J. B., et al. (2019). Ice nucleating particles in the marine boundary layer in the Canadian Arctic during summer 2014. Atmospheric Chemistry and Physics, 19(2), 1027–1039. https://doi.org/10.5194/ acp-19-1027-2019
- Jahn, L. G., Polen, M. J., Jahl, L. G., Brubaker, T. A., Somers, J., & Sullivan, R. C. (2020). Biomass combustion produces ice-active minerals in biomass-burning aerosol and bottom ash, *Proceedings of the National Academy of Sciences of the United States of America*. (Vol. 117(36), pp. 21928–21937). https://doi.org/10.1073/pnas.1922128117
- Juhls, B., Overduin, P. P., Hölemann, J., Hieronymi, M., Matsuoka, A., Heim, B., & Fischer, J. (2019). Dissolved organic matter at the fluvialmarine transition in the Laptev Sea using in situ data and ocean color remote sensing. *Biogeosciences*, 16(13), 2693–2713. https://doi. org/10.5194/bg-16-2693-2019

Juhls, B., Stedmon, C. A., Morgenstern, A., Meyer, H., Hölemann, J., Heim, B., et al. (2020). Identifying drivers of seasonality in lena river biogeochemistry and dissolved Organic Matter Fluxes. Frontiers in Environmental Science, 8(53). https://doi.org/10.3389/fenvs.2020.00053

Kanji, Z. A., Ladino, L. A., Wex, H., Boose, Y., Burkert-Kohn, M., Cziczo, D. J., & Krämer, M. (2017). Overview of ice nucleating particles. *Meteorological Monographs*, 58(33), 1–11. https://doi.org/10.1175/amsmonographs-d-16-0006.1

Karlsson, L., & Zieger, P. (2020). Aerosol particle number size distribution data collected during the Arctic Ocean 2018 expedition. Bolin Centre Database. Dataset version 1. https://doi.org/10.17043/oden-ao-2018-aerosol-dmps-1

- Kay, J. E., & L'Ecuyer, T. (2013). Observational constraints on Arctic Ocean clouds and radiative fluxes during the early 21st century. Journal of Geophysical Research, 118(13), 7219–7236. https://doi.org/10.1002/jgrd.50489
- Kerminen, V. M., & Leck, C. (2001). Sulfur chemistry over the central Arctic Ocean during the summer: Gas-to-particle transformation. Journal of Geophysical Research, 106(D23), 32087–32099. https://doi.org/10.1029/2000jd900604
- Khan, M. A. H., Gillespie, S. M. P., Razis, B., Xiao, P., Davies-Coleman, M. T., Percival, C. J., et al. (2016). A modelling study of the atmospheric chemistry of DMS using the global model, STOCHEM-CRI. Atmospheric Environment, 127, 69–79. https://doi.org/10.1016/j. atmosenv.2015.12.028
- Knackstedt, K. A., Moffett, B. F., Hartmann, S., Wex, H., Hill, T. C. J., Glasgo, E. D., et al. (2018). Terrestrial origin for abundant riverine nanoscale ice-nucleating particles. *Environmental Science & Technology*, 52(21), 12358–12367. https://doi.org/10.1021/acs.est.8b03881
- Kupiszewski, P., Leck, C., Tjernström, M., Sjogren, S., Sedlar, J., Graus, M., et al. (2013). Vertical profiling of aerosol particles and trace gases over the central Arctic Ocean during summer. Atmospheric Chemistry and Physics, 13(24), 12405–12431. https://doi.org/10.5194/ acp-13-12405-2013
- Larsen, J. A., Conen, F., & Alewell, C. (2017). Export of ice nucleating particles from a watershed, Royal Society Open Science, 4(8), 170213. https://doi.org/10.1098/rsos.170213
- Lawler, M. J., Draper, D. C., & Smith, J. N. (2020). Atmospheric fungal nanoparticle burstsJ Science Advances, 6(3), eaax9051. https://doi. org/10.1126/sciadv.aax9051%
- Leck, C., & Bigg, E. K. (2005). Biogenic particles in the surface microlayer and overlaying atmosphere in the central Arctic Ocean during summer. *Tellus Series B Chemical and Physical Meteorology*, 57(4), 305–316. https://doi.org/10.3402/tellusb.v57i4.16546
- Leck, C., & Persson, C. (1996). Seasonal and short-term variability in dimethyl sulfide, sulfur dioxide and biogenic sulfur and sea salt aerosol particles in the arctic marine boundary layer during summer and autumn. *Tellus Series B Chemical and Physical Meteorology*, 48(2), 272–299. https://doi.org/10.1034/j.1600-0889.48.issue2.1.x

Daily, M. I., Whale, T., Tarn, M. D., & M.B.J (2021).

- Leck, C., Matrai, P., Achtert, P., Adams, M., Baccarini, A., Brooks, B., et al. (2018). Data from expedition Arctic Ocean [Data set]. Bolin Centre Database. https://doi.org/10.17043/oden-ao-2018-expedition-1
- Leck, C., & Svensson, E. (2015). Importance of aerosol composition and mixing state for cloud droplet activation over the Arctic pack ice in summer. Atmospheric Chemistry and Physics, 15(5), 2545–2568. https://doi.org/10.5194/acp-15-2545-2015
- Lohmann, U., & Leck, C. (2005). Importance of submicron surface-active organic aerosols for pristine Arctic clouds. Tellus Series B Chemical and Physical Meteorology, 57(3), 261–268. https://doi.org/10.3402/tellusb.v57i3.16534
- Maslanik, J., & Stroeve, J. (1999). In Near-real-time DMSP SSMIS daily polar gridded sea ice ConcentrationsNASA National Snow and Ice Data Center distributed active. Archive Center. Version 1 https://doi.org/10.5067/U8C09DWVX9LM
- Mason, R. H., Si, M., Chou, C., Irish, V. E., Dickie, R., Elizondo, P., et al. (2016). Size-resolved measurements of ice-nucleating particles at six locations in North America and one in Europe. *Atmospheric Chemistry and Physics*, 16(3), 1637–1651. https://doi.org/10.5194/ acp-16-1637-2016
- McCluskey, C. S., Hill, T. C. J., Humphries, R. S., Rauker, A. M., Moreau, S., Strutton, P. G., et al. (2018). Observations of ice nucleating particles over Southern Ocean waters. *Geophysical Research Letters*, 45(2111), 11989–11997. https://doi.org/10.1029/2018g1079981
- McCluskey, C. S., Ovadnevaite, J., Rinaldi, M., Atkinson, J., Belosi, F., Ceburnis, D., et al. (2018). Marine and terrestrial organic ice-nucleating particles in pristine marine to continentally influenced Northeast atlantic air masses. *Journal of Geophysical Research*, 123(11), 6196–6212. https://doi.org/10.1029/2017JD028033
- Meinander, O., Waldhauserova, P. D., Amosov, P., Aseyeva, E., Atkins, C., Kandler, K., et al. (2021). Newly identified climatically and environmentally significant high latitude dust sources. Atmospheric Chemistry and Physics Discussions, 2021, 1–74. https://doi.org/10.5194/ acp-2021-963
- Morrison, H., de Boer, G., Feingold, G., Harrington, J., Shupe, M. D., & Sulia, K. (2012). Resilience of persistent Arctic mixed-phase clouds. *Nature Geoscience*, 5(1), 11–17. https://doi.org/10.1038/ngeo1332
- Murray, B. J., Carslaw, K. S., & Field, P. R. (2021). Opinion: Cloud-phase climate feedback and the importance of ice-nucleating particles. Atmospheric Chemistry and Physics, 21(2), 665–679. https://doi.org/10.5194/acp-21-665-2021
- Murray, B. J., O'Sullivan, D., Atkinson, J. D., & Webb, M. E. (2012). Ice nucleation by particles immersed in supercooled cloud droplets. *Chem*ical Society Reviews, 41(19), 6519–6554. https://doi.org/10.1039/c2cs35200a
- O'Dowd, C. D., Facchini, M. C., Cavalli, F., Ceburnis, D., Mircea, M., Decesari, S., et al. (2004). Biogenically driven organic contribution to marine aerosol. *Nature*, 431(7009), 676–680. https://doi.org/10.1038/nature02959
- Orellana, M. V., Matrai, P. A., Leck, C., Rauschenberg, C. D., Lee, A. M., & Coz, E. (2011). Marine microgels as a source of cloud condensation nuclei in the high Arctic. Proceedings of the National Academy of Sciences of the United States of America, 108(33), 13612–13617. https:// doi.org/10.1073/pnas.1102457108
- O'Sullivan, D., Adams, M. P., Tarn, M. D., Harrison, A. D., Vergara-Temprado, J., Porter, G. C. E., et al. (2018). Contributions of biogenic material to the atmospheric ice-nucleating particle population in North Western Europe. *Scientific Reports*, 8(1), 13821. https://doi.org/10.1038/ s41598-018-31981-7
- O'Sullivan, D., Murray, B. J., Ross, J. F., & Webb, M. E. (2016). The adsorption of fungal ice-nucleating proteins on mineral dusts: A terrestrial reservoir of atmospheric ice-nucleating particles. *Atmospheric Chemistry and Physics*, 16(12), 7879–7887. https://doi.org/10.5194/ acp-16-7879-2016
- O'Sullivan, D., Murray, B. J., Ross, J. F., Whale, T. F., Price, H. C., Atkinson, J. D., et al. (2015). The relevance of nanoscale biological fragments for ice nucleation in clouds. *Scientific Reports*, 5, 8082. https://doi.org/10.1038/srep08082
- Petters, M. D., & Wright, T. P. (2015). Revisiting ice nucleation from precipitation samples. Geophysical Research Letters, 42(20), 8758–8766. https://doi.org/10.1002/2015GL065733
- Porter, G. C. E., Murray, B. J., Adams, A. M., Brooks, I. M., Zieger, P., & Karlsson, L. (2022). Ice-Nucleating particle data from the cruise to the North Pole on board the Oden Icebreaker [Data set]. University of Leeds. https://doi.org/10.5518/1093
- Porter, G. C. E., Sikora, S. N. F., Adams, M. P., Proske, U., Harrison, A. D., Tarn, M. D., et al. (2020). Resolving the size of ice-nucleating particles with a balloon deployable aerosol sampler: The SHARK. Atmospheric Measurement Techniques, 13(6), 2905–2921. https://doi. org/10.5194/amt-13-2905-2020
- Prenni, A. J., Harrington, J. Y., Tjernstrom, M., DeMott, P. J., Avramov, A., Long, C. N., et al. (2007). Can ice-nucleating aerosols affect arctic seasonal climate? *Bulletin of the American Meteorological Society*, 88(4), 541–550. https://doi.org/10.1175/bams-88-4-541
- Pummer, B. G., Budke, C., Augustin-Bauditz, S., Niedermeier, D., Felgitsch, L., Kampf, C. J., et al. (2015). Ice nucleation by water-soluble macromolecules. Atmospheric Chemistry and Physics, 15(8), 4077–4091. https://doi.org/10.5194/acp-15-4077-2015
- Reicher, N., Segev, L., & Rudich, Y. (2018). The WeIzmann Supercooled Droplets Observation on a Microarray (WISDOM) and application for ambient dust. Atmospheric Measurement Techniques, 11(1), 233–248. https://doi.org/10.5194/amt-11-233-2018
- Rogers, D. C., DeMott, P. J., & Kreidenweis, S. M. (2001). Airborne measurements of tropospheric ice-nucleating aerosol particles in the Arctic spring. *Journal of Geophysical Research*, 106(D14), 15053–15063. https://doi.org/10.1029/2000jd900790
- Sanchez-Marroquin, A., Arnalds, O., Baustian-Dorsi, K. J., Browse, J., Dagsson-Waldhauserova, P., Harrison, A. D., et al. (2020). Iceland is an episodic source of atmospheric ice-nucleating particles relevant for mixed-phase clouds. *Science Advances*, 6(26), eaba8137. https://doi. org/10.1126/sciadv.aba8137
- Sanchez-Marroquin, A., West, J. S., Burke, I., McQuaid, J. B., & Murray, B. J. (2021). Mineral and biological ice-nucleating particles above the south East of the British isles. *Environmental Sciences: Atmosphere*, 1, 176–191. https://doi.org/10.1039/D1EA00003A
- Schill, G. P., DeMott, P. J., Emerson, E. W., Rauker, A. M. C., Kodros, J. K., Suski, K. J., et al. (2020). The contribution of black carbon to global ice nucleating particle concentrations relevant to mixed-phase clouds. *Proceedings of the National Academy of Sciences of the United States of America*, 117(37), 22705–22711. https://doi.org/10.1073/pnas.2001674117
- Schmale, J., Zieger, P., & Ekman, A. M. L. (2021). Aerosols in current and future Arctic climate. Nature Climate Change, 11(2), 95–105. https:// doi.org/10.1038/s41558-020-00969-5
- Schneider, J., Höhler, K., Heikkilä, P., Keskinen, J., Bertozzi, B., Bogert, P., et al. (2021). The seasonal cycle of ice-nucleating particles linked to the abundance of biogenic aerosol in boreal forests. *Atmospheric Chemistry and Physics*, 21(5), 3899–3918. https://doi.org/10.5194/ acp-21-3899-2021

Schnell, R. C., & Vali, G. (1975). Freezing nuclei in marine waters. *Tellus*, 27(3), 321–323. https://doi.org/10.3402/tellusa.v27i3.9911

Shupe, M. D., Persson, P. O. G., Brooks, I. M., Tjernstrom, M., Sedlar, J., Mauritsen, T., et al. (2013). Cloud and boundary layer interactions over the Arctic sea ice in late summer. Atmospheric Chemistry and Physics, 13(18), 9379–9399. https://doi.org/10.5194/acp-13-9379-2013

Si, M., Evoy, E., Yun, J., Xi, Y., Hanna, S. J., Chivulescu, A., et al. (2019). Concentrations, composition, and sources of ice-nucleating particles in the Canadian High Arctic during spring 2016. Atmospheric Chemistry and Physics, 19(5), 3007–3024. https://doi.org/10.5194/acp-19-3007-2019

- Sprenger, M., & Wernli, H. (2015). The LAGRANTO Lagrangian analysis tool Version 2.0. Geoscientific Model Development, 8(8), 2569–2586. https://doi.org/10.5194/gmd-8-2569-2015
- Stevens, R. G., Loewe, K., Dearden, C., Dimitrelos, A., Possner, A., Eirund, G. K., et al. (2018). A model intercomparison of CCN-limited tenuous clouds in the high Arctic. Atmospheric Chemistry and Physics, 18(15), 11041–11071. https://doi.org/10.5194/acp-18-11041-2018
- Stohl, A., Klimont, Z., Eckhardt, S., Kupiainen, K., Shevchenko, V. P., Kopeikin, V. M., & Novigatsky, A. N. (2013). Black carbon in the arctic: The underestimated role of gas flaring and residential combustion emissions. *Atmospheric Chemistry and Physics*, 13(17), 8833–8855. https:// doi.org/10.5194/acp-13-8833-2013
- Tan, I., & Storelvmo, T. (2019). Evidence of strong contributions from mixed-phase clouds to arctic climate change. Geophysical Research Letters, 46(5), 2894–2902. https://doi.org/10.1029/2018GL081871
- Thomson, E. S., Weber, D., Bingemer, H. G., Tuomi, J., Ebert, M., & Pettersson, J. B. C. (2018). Intensification of ice nucleation observed in ocean ship emissions. *Scientific Reports*, 8(1), 1111. https://doi.org/10.1038/s41598-018-19297-y
- Tjernstrom, M., Birch, C. E., Brooks, I. M., Shupe, M. D., Persson, P. O. G., Sedlar, J., et al. (2012). Meteorological conditions in the central arctic summer during the arctic summer cloud ocean study (ASCOS). Atmospheric Chemistry and Physics, 12(15), 6863–6889. https://doi. org/10.5194/acp-12-6863-2012
- Tobo, Y., Adachi, K., DeMott, P. J., Hill, T. C. J., Hamilton, D. S., Mahowald, N. M., et al. (2019). Glacially sourced dust as a potentially significant source of ice nucleating particles. *Nature Geoscience*, *12*(4), 253–258. https://doi.org/10.1038/s41561-019-0314-x
- Ullrich, R., Hoose, C., Möhler, O., Niemand, M., Wagner, R., Höhler, K., et al. (2017). A new ice nucleation active site parameterization for desert dust and soot. Journal of the Atmospheric Sciences, 74(3), 699–717. https://doi.org/10.1175/jas-d-16-0074.1
- Umo, N. S., Murray, B. J., Baeza-Romero, M. T., Jones, J. M., Lea-Langton, A. R., Malkin, T. L., et al. (2015). Ice nucleation by combustion ash particles at conditions relevant to mixed-phase clouds. *Atmospheric Chemistry and Physics*, 15(9), 5195–5210. https://doi.org/10.5194/ acp-15-5195-2015
- Vassel, M., Ickes, L., Maturilli, M., & Hoose, C. (2019). Classification of Arctic multilayer clouds using radiosonde and radar data in Svalbard. Atmospheric Chemistry and Physics, 19(7), 5111–5126. https://doi.org/10.5194/acp-19-5111-2019
- Vergara-Temprado, J., Holden, M. A., Orton, T. R., O'Sullivan, D., Umo, N. S., Browse, J., et al. (2018). Is black carbon an unimportant ice-nucleating particle in mixed-phase clouds? *Journal of Geophysical Research*, 123(8), 4273–4283. https://doi.org/10.1002/2017JD027831
- Vergara-Temprado, J., Miltenberger, A. K., Furtado, K., Grosvenor, D. P., Shipway, B. J., Hill, A. A., et al. (2018). Strong control of Southern Ocean cloud reflectivity by ice-nucleating particles. *Proceedings of the National Academy of Sciences of the United States of America*, 115, 2687–2692. https://doi.org/10.1073/pnas.1721627115
- Vergara-Temprado, J., Murray, B. J., Wilson, T. W., O'Sullivan, D., Browse, J., Pringle, K. J., et al. (2017). Contribution of feldspar and marine organic aerosols to global ice nucleating particle concentrations. *Atmospheric Chemistry and Physics*, 17(5), 3637–3658. https://doi. org/10.5194/acp-17-3637-2017
- Vüllers, J., Achtert, P., Brooks, I. M., Tjernström, M., Prytherch, J., Burzik, A., & Neely, R., III. (2021). Meteorological and cloud conditions during the Arctic Ocean 2018 expedition. Atmospheric Chemistry and Physics, 21(1), 289–314. https://doi.org/10.5194/acp-21-289-2021
- Wagner, R., Jähn, M., & Schepanski, K. (2018). Wildfires as a source of airborne mineral dust Revisiting a conceptual model using large-eddy simulation (LES). Atmospheric Chemistry and Physics, 18(16), 11863–11884. https://doi.org/10.5194/acp-18-11863-2018
- Welti, A., Bigg, E. K., DeMott, P. J., Gong, X., Hartmann, M., Harvey, M., et al. (2020). Ship-based measurements of ice nuclei concentrations over the Arctic, Atlantic, Pacific and Southern oceans. Atmospheric Chemistry and Physics, 20(23), 15191–15206. https://doi.org/10.5194/ acp-20-15191-2020
- Wex, H., Huang, L., Zhang, W., Hung, H., Traversi, R., Becagli, S., et al. (2019). Annual variability of ice-nucleating particle concentrations at different Arctic locations. Atmospheric Chemistry and Physics, 19(7), 5293–5311. https://doi.org/10.5194/acp-19-5293-2019
- Whale, T. F., Murray, B. J., O'Sullivan, D., Wilson, T. W., Umo, N. S., Baustian, K. J., et al. (2015). A technique for quantifying heterogeneous ice nucleation in microlitre supercooled water droplets. *Atmospheric Measurement Techniques*, 8(6), 2437–2447. https://doi.org/10.5194/ amt-8-2437-2015
- Wilson, T. W., Ladino, L. A., Alpert, P. A., Breckels, M. N., Brooks, I. M., Browse, J., et al. (2015). A marine biogenic source of atmospheric ice-nucleating particles. *Nature*, 525(7568), 234–238. https://doi.org/10.1038/nature14986
- Zieger, P., & Karlsson, L. (2022a). Coarse-mode particle number size distribution data collected during the Arctic Ocean 2018 expedition [Data set version 1]. Bolin Centre Database. https://doi.org/10.17043/oden-ao-2018-aerosol-coarse-1
- Zieger, P., & Karlsson, L. (2022b). Equivalent black carbon concentration measured during the Arctic Ocean 2018 expedition [Data set version 1]. Bolin Centre Database. https://doi.org/10.17043/oden-ao-2018-aerosol-ebc-1